Author's response to reviews

Title: Stroke with atrial fibrillation or atrial flutter: descriptive population-based study from the Brest Stroke Registry.

Authors:

Virginie Jannou (virginie.jannou@chu-brest.fr)
Serge Timsit (serge.timsit@chu-brest.fr)
Emmanuel Nowak (emmanuel.nowak@chu-brest.fr)
François Rouhart (francois.rouhart@chu-brest.fr)
Philippe Goas (philippe.goas@chu-brest.fr)
François-Mathias Merrien (francois-mathia.merrien@chu-brest.fr)
Irina Viakhireva-Dovganyuk (irina.viakhireva@chu-brest.fr)
Anne Tirel-Badets (anne.tirel.badets@gmail.com)
Armelle Gentric (armelle.gentric@chu-brest.fr)

Version: 2 Date: 25 March 2015

Author's response to reviews:

Brest, March 25th, 2015

Dear Prof Flicker,

Dear Dr Feigin,

Please find attached the answers to comments from reviewers for the article entitled “Stroke with atrial fibrillation or atrial flutter: descriptive population-based study from the Brest Stroke Registry” from Jannou et al.

We have tried to answer to all the comments and modified the text accordingly.

For clarity, you will find three documents:
- One with the answers to the reviewers
- One article revised in yellow for text which is added to the article
- One article completely modified with no apparent correction (yellow)

I hope that we have taken in account all comments.

Sincerely Yours,

Pr S. Timsit